

St. MARY'S UNIVERSITY SCHOOL OF GRADUATE STUDIES DEPARTMENT OF BUSINESS ADMINISTRATION

CHALLENGES TO THE SUCCESSFUL IMPLEMEANTATION OF BALANCED SCORE CARD AS PERFORMANCE MANAGEMENT SYSTEM-THE CASE OF ALERT HOSPITAL

By: -MIHRET HUSSIEN

DECEMBER 2020, SMU ADDIS ABABA

CHALLENGES TO THE SUCCESSFUL IMPLEMEANTATION OF BALANCED SCORE CARD AS PERFORMANCE MANAGEMENT SYSTEM-THE CASE OF ALERT HOSPITAL

By: -MIHRET HUSSIEN

ADVISOR: MELAKU GIRMA (PhD)

A THESIS SUBMITTED TO THE SCHOOL OF GRADUATE STUDIES
OF SMU PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE
DEGREE OF MASTER OF BUSINESS ADMINISTRATION

TABLE OF CONTENT

Abs	tract	i
Ack	nowledgments	ii
List	of Tables/Figures	iii
List	of Acronyms and/or Abbreviations	iv
Boa	rd of Examiners	v
Dec	laration	vi
CH	APTER ONE: INTRODUCTION	01
1.1	Background of the Study	
1.2	Statement of the Problem	03
1.3	Basic Research Questions	05
1.4	Objective of the Study	05
	1.4.1 General Objective	05
	1.4.2 Specific Objective	05
1.5	Significance of the Study	05
1.6	Delimitation/Scope of the Study	06
1.7	Limitations of the Study	06
1.8	Definition of Terms	06
1.9	Organization of the Study	07
CH	APTER TWO: REVIEW OF RELATED LITERATURE	08
2.1	Theoretical Literature Review	08
	2.1.1 Brief Overview of Performance Measurement	08
	2.1.2 The Origins of the Balanced Scorecard	09
	2.1.3 The Concept of Balanced Scorecard	11
	2.1.4 Balanced Scorecard Perspectives	12

	2.1.4.1	Financial Perspective	13
	2.1.4.2	Customer Perspective	13
	2.1.4.3	Internal Business Process Perspective	14
	2.1.4.4	Learning and Growth Perspective	15
	2.1.5 The Step	ps of Building and Implementing Balanced Scorecard	16
	2.1.6 Succes	s Factors of Balanced Scorecard Implementation	18
2.2	Empirical Lite	erature Review	19
	2.2.1 Brief Ov	verview of BSC in Africa	19
	2.2.2 Brief Ba	ackground of BSC in Ethiopia	20
	2.2.3 Signification	ance of BSC in the Health Sector	21
2.3	Conceptual Fr	ame Work	23
СН	APTER THE	REE: RESEARCH DESIGN and METHODOLOGY	27
3.1	Research Mo	ethodology	27
3.2		esign	
3.3		S	
3.4	Data Collec	tion Techniques	28
3.5	Study Area	and Population	29
3.6	Sample and	Sampling Procedures	29
3.7	Data Analy	sis Methods	30
3.8	Reliability		31
3.9	Validity		31
3.10	Ethical Cons	sideration	32
СН	APTER FOU	JR: DATA ANALYSIS AND INTERPRETATION	33
4.1		's Profile	
4.2	Analysis of	the Variables	34
	4.2.1 Concep	pt Clarity of the Balanced Scorecard	35
	4.2.2 Execut	ive Sponsorship	36
	4.2.3 Balanc	ed scorecard education and training	_37

	4.2.4 IT Support	38
	4.2.5 Competent Project Team	39
	4.2.6 Organizational Level Participation	40
	4.2.7 Key Performance Indicators	41
	4.2.8 Planning and Communication	41
4.3	Analysis of the Interview	41
CH	APTER FIVE: SUMMARY, CONCLUSIONS AND RECOMMENDATIONS	44
5.1	Summary of the Major Findings	44
5.2	Conclusion	45
5.3	Recommendations	46
Ref	erence	48
App	oendices	53
App	endix I: Questionnaire	
App	endix II: Interview Guide	

ABSTRACT

The main purpose of the study was to assess the overall implementation and challenges faced while implementing BSC as a tool of performance management system in the case of All Africa Leprosy, Tuberculosis, Training and Rehabilitation Center/ALERT/. The assessment attempted to examine the extent in the effectiveness of BSC planning, its implementation and its achievement, identifying major challenge encountered by ALERT and mechanism or strategies in place to ease the implementation. Descriptive survey method was employed to conduct this study. With regard to sampling the respondent of the study were conveniently stratified by the organization as management and nonmanagement staff that was major source of the data. Then 260 management and non-management staffs were selected by those who are with Diploma and above sampling techniques. To attain the objectives, the study were employed both quantitative & qualitative research approaches. The entire closed end questions were analyzed quantitatively using the Frequency, Percentages, Average Mean, Standard Besides, the data obtained from closed-ended questions, interviews, observation and deviation, document analysis were transcribed to supplement the quantitative data. In order to assess its effectiveness in performance management, the study utilized the four perspectives of the BSC. Based on these the data was computed in SPSS and analyzed. The finding revealed that, majority of the common implementation challenges such as limited understanding of BSC, lack of executive sponsorship, lack of BSC education and training, inadequate IT support, inadequate project team and organizational participation, inadequate key performance indicators (KPIs) and lack of planning and communication were observed in the stated organization at different level. However, lack of formal BSC education and training, lack of planning and communication, lack of organizational participation and inadequate IT support were the major challenges identified. The study advised the organization to conduct intensive awareness creation activities, ensure organizational level participation, develop strong implementation and monitoring mechanism and back the system with relevant IT system to effectively take advantage of the Balanced Scorecard system.

Key words: Balanced scorecard, Performance measurement

ACKNOWLEDGEMENTS

First and foremost, I would like to thank my Lord to help me in all my endeavors, and to be with me in times of finalize my research. Next I am thankful to my advisor Dr. Melaku Girma for his suggestion, comments, and follow up. Besides, I am very thankful to my family, my father & Mother and for their understanding, courage and inspiration.

Further, I would like to thank all the employees of All Africa Leprosy, Tuberculosis, Training and Rehabilitation Center/ALERT/ for their positive approach and cooperation.

Lastly I would like to extend my immense thankfulness to my parents and friends and to all who have even a single contribution in this work.

List of tables/Figures

P	a	g	e
_		$\overline{}$	_

Figure 1: Balanced Scorecard Framework/The Four Perspectives of BSC/	15
Figure 3: Conceptual Framework	26
Table 4.1.1 Personal Information of Respondents	33
Tables 4.2.1 Concept Clarity of the Balanced Scorecard	35
Tables 4.2.2 Level of Executive Sponsorship	36
Tables-4.2.3 Balanced scorecard education and training	37
Tables-4.2.4 BSC Automation	38
Table 4.2.5 Effectiveness of Dedicated Project Team	39
Table 4.2.6: Organizational Level Participation	40
Table 4.2.7: Key Performance Indicators	41
Table 4.2.8: Planning and Communication of BSC System	42

List of Acronyms and/or Abbreviations

ALERT All Africa Leprosy, Tuberculosis, Training and Rehabilitation Center

BSC Balanced Score Card

BPR Business Process Reengineering

CSRP Civil Service Reform Program

FMOH Federal Ministry of Health

GoE Government of Ethiopia

IT Information Technology

MBO Management by objective

MLI Ministerial Leadership Initiative for Global Health

MoCB Ministry of Capacity Building

MoCS Ministry of Civil Service

MoFED Ministry of Finance and Economic Development

NPLs None Performing Loans

PEST Political, Economic, Social and Technological

PSCBP Public Sector Capacity Building Program

SWOT Strength, Weakness, Opportunity, and Threat

SPM Strategic Planning and Management

SPSS Statistical Package for Social Science

St. MAR'S UNIVERSITY SCHOOL OF GRADUATE STUDIES FACULTY OF BUSINESS

CHALLENGES TO THE SUCCESSFUL IMPLEMEANTATION OF BALANCED SCORE CARD AS PERFORMANCE MANAGEMENT SYSTEM-THE CASE OF ALERT HOSPITAL

BY: MIHRET HUSSIEN

APPROVED BY: BOARD OF EXAMINERS

Advisor	Signature
Melaku Girma (PhD)	_hantle.
Dean, Graduate Studies	Signature
External Examiner	Signature
Dr. Getie Andualem	Setand
Internal Examiner	Signature
Mohammed Mohammednur (Asst. Prof.)	- Hyline

DECLARATION

I hereby declare that this submission is my own work towards the Masters of Business Administration prepared under the guidance of Melaku Girma (PhD) and that, to the best of my knowledge, it contains neither material previously published by another person nor material which has been accepted for the award of any other degree of the university, except where due acknowledgment has been made in the text.

Mihret Hussien SGS/0088/2011A

Signature

CHAPTER ONE

INTRODUCTION

The focus of this chapter for the overall study gives an outline specifically stating the statement of the problem which motivated the researcher combined with research questions. The chapter also delimits the scope and significance of the study.

1.1 Background of the Study

Organizations in today's change filled, highly competitive environment must give significant time, energy, and human and financial resources to measuring their performance in achieving strategic goals. Most do just that, but despite the substantial effort and related costs, many are dissatisfied with their measurement efforts (Niven, 2006).

Performance measurement system enables informed decisions to be made and actions to be taken because it quantifies the efficiency and effectiveness of past actions through the acquisition, collation, sorting, analysis and interpretation of appropriate data (Kermally, 1997).

It is clear that the rules of the game for business have changed materially over the last decade. Or, at the very least, the rules of the game are in the process of radical change. Success for organizations today is measured very differently than it was yesterday. Of course financial performance is still essential. Delivering profit growth or enhancing shareholder value is still at the top of every executive's agenda, but it is now widely recognized that delivering financial performance alone is insufficient. Even more importantly, it is now generally accepted that the level of financial performance achieved today is a function of decisions made 6–18 months or even longer (Kermally, 1997).

According to Norton and Kaplan (1992) during 1980s, many executives were convinced that traditional measures of financial performance didn't let them manage effectively and wanted to replace them with operational measures. Arguing that executives should track both financial and operational metrics, Norton and Kaplan suggested four sets of parameters, which finally leads to the birth of the Balanced Scorecard.

The Balanced Scorecard is a set of measures that gives top managers a fast but comprehensive view of the business. In addition to the traditional financial measures that tell the results of actions already taken, the balanced scorecard accompanies other operational measures on customer

satisfaction, internal processes, and the organization's innovation measures that are drivers of future financial performance (Norton and Kaplan, 1992).

In addition to this, these authors have found out that apart from being a tool of performance management, the BSC translates an organization's mission and strategy into a comprehensive set of performance measures that provides the framework for a strategic measurement and management system (Norton and Kaplan, 1996).

Regardless of these benefits that are promised by the balanced scorecard, empirical studies show that there are challenges of implementation. According to Pujas (2010) common challenges of implementation of BSC are limited understanding of BSC, lack of executive sponsorship, lack of BSC education and training, inadequate IT support, inadequate project team, not involving the whole organization, inadequate key performance indicators (KPIs) and lack of planning and communication.

The Ethiopian public sector organizations had been trying to implement various modern management systems and tools since 1995 such as, "Result-oriented Performance Evaluation Method", Business Process Reengineering (BPR), Balanced Scorecard (BSC) and kaizen in order to improve efficiency (Adebabay, 2011).

In 2007, the FMOH boarded on a journey to transform its strategic planning and management processes by adopting the Balanced Scorecard (BSC) as their framework. Extensive work has been done at the leadership level to inculcate the sector with the concepts of balanced scorecard and to link day-to-day activities with the organization's long term direction. (Zemetis, 2009).

In cognizance of the salience of this tool, former Minister has applied the BSC approach as director of the Tigray Regional Health Bureau. The Federal Ministry of Health (FMOH) was convinced that with adequate adaptation, the BSC approach could deliver significantly improved performance throughout the health system. In 2007 the FMOH became the first ministerial office of the Government of Ethiopia (GoE) to apply the BSC approach, although it is the intent of the ministry of capacity building that BSC approach be used government wide (Ethiopia Balanced Scorecard Overview, 2010).

The Ethiopia's Health Sector regulatory organizations exist at various tiers. In other words, the structure of the sector is made of the Federal Ministry of Health (FMOH), Federal Health

Directorates and Agencies, Regional Health Bureaus, Woreda Zone offices, Woredas (an administrative district unit), and individual health facilities including federal and regional hospitals, health centers, and health posts.

Therefore, this study assessed the overall implementation of the balanced scorecard system as a performance measurement, strategic management and communication system in the health center of Ethiopia, at All Africa Leprosy, Tuberculosis, Training and Rehabilitation Center/ALERT/. Currently the center has 1,342 permanent and 40 contract total 1,382 employees.

1.2 Statement of the Problem

The Ethiopian government has given high emphasis and dedication towards institutional excellence; building on earlier efforts to establish strategic planning norms and an employee result-oriented performance appraisal system, the federal government introduced the BSC in 2010 (Menwuyelet Fentie, 2014).

At present, almost all public institutions are undertaking Balanced Scorecard (BSC) as a strategic planning and management tools to measure work performance of any organization. Among the effects for public sector reform is the efforts to improve public sector services, through process improvement. The reforming aimed at promotion of efficiency (cost effectiveness) and provision of effective service to the public. In addition many large companies use a performance measurement system. Companies that start with a performance measurement system face difficulties with implementation Molleman (2007). Adaption of balanced score card would be beneficial but the implementation would be challenging (Stewart & Carpenter-Hubin, 2001, cited by Hiwot T. 2015).

Much of the empirical literature on BSC concerns, it is estimated that 70% of BSC implementation has failed Atkinson (2006). Dragan (2010) on his thesis conclude that although many different organizations are using the BSC as a management technique to implement corporate strategy, a number of them have encountered different problems when trying to introduce the BSC and implement it. The majority have either implemented the BSC but without any significant improvement in performance, or they have given up in the implementation the process itself.

Numerous organizations develop good strategies, develop good action plans, consider organizational structure, take a close look at their human resource needs, fund their strategies through their annual business plan, and develop a plan to monitor and control their strategies and tactics. And yet they still fail to successfully implement those strategies and tactics. The reason, most often, is they lack linkage (Birnbaum, B., 2007, cited by Hiwot T. 2015). Linkage involves tying together all the activities of the organization to make sure that all of the organizational resources are rowing in the same direction. Problems often encountered in strategic plans include lack of accuracy of the plans, inconsistency in objectives and inadequate rewards for efforts expended in the planning process. The leadership style to be followed, leader's and manager's commitment, measurement error in performance, employee motivation and communication problems are also the main challenges to BSC implementation for successful organizational performance.

Many of the studies focus on the motivation for adopting BSC, nature of performance indicators, and the perceive benefits. However, studies on the implementation of BSC in public sector organizations are erratic. Moreover, though there are some evidences about the achievements and challenges of implementing the BSC at the national level, from the informal interviews conducted and personal observation, still little is known about the achievements and challenges of the change initiatives at the ALERT Hospital.

This study, therefore, seeks to assess the extent of BSC implementation, its success and challenges and will recommend appropriate solutions to the problems faced in the implementation in the ALERT Hospital.

1.3 Basic Research Questions

Based on the statement of the problem the researcher will try to seek answer for the following three research questions.

- How is Balanced Scorecard being implemented in ALERT compared to the standard literature?
- What are the major achievements gained and challenges confronted in the implementation of BSC?

What are the key factors that influence effective implementation of BSC in this organization?

1.4 Objective of the Study

1.4.1 General Objective

The general objective of the study is to examine the overall practices of implementing balanced scorecard in ALERT and sort out challenges faced during implementation.

1.4.2 Specific Objective

The specific objectives of the study will be to:

- To assess the challenges of implementing BSC in the ALERT.
- To evaluate how the balanced scorecard is implemented in the ALERT and evaluate the key success factors for effective implementation.
- To reveal possible recommendations for the implementation challenges identified.

1.5 Significance of the Study

The following are the major significances of the study:

- It will serve as an input for the process owners and employees in ALERT to re-examine how much they were effective in implementing BSC towards improving organizational performance.
- It provides important information about the main challenges and problems encountered in the implementation of balanced scorecard with possible suggestions to take corrective actions in the future.
- It will serve as a benchmark for those organizations who did not fully implement balanced scorecard.
- It helps the researcher to acquire knowledge and practical experience about balanced scorecard.
- Moreover, the researcher also believes that this study can potentially serve as a stepping stone for the future research in this area.

1.6 Scope of the study

The scope of the study was bounded to assessing the implementation gaps and challenges faced by the case of All Africa Leprosy, Tuberculosis, Training and Rehabilitation Center/ALERT/.

The study conducted the survey on employees and management of the stated organization which is found in Addis Ababa, Ethiopia. The study assessed the implementation status of the system till May 31, 2020.

1.7 Limitations of the Study

The major limitation of the study was lack of relevant empirical literature. Lack of cooperation among some individuals and organizations in providing data required for the study. Some of the questionnaires were not returned back due to different reasons. Furthermore, shortage of accessibility of relevant documents that clearly shows the overall implementation of the balanced scorecard system in the organization was also among the major challenges.

1.8 Definition of Terms

<u>Performance measurement</u> is a complex issue that normally incorporates at least four disciplines: economics, management, accounting and information technology (Tagen, 2004).

Balanced Scorecard: is a management tool that enables an organization to translate its vision and strategy into a tangible set of performance measures. However, it is more than a measuring device (Kaplan and Norton, 1996).

<u>Balanced Score Card Perspectives</u>- refers to category of performance objectives or measures in is in balanced scorecard and in most organization choose the four standard perspectives such as; financial, customer, internal process, learning and growth (Kaplan and Norton, 1996).

<u>Cascade:</u> The process of developing aligned scorecards throughout an organization each level of the organization will develop scorecards based on the objectives and measure. It can influence from the scorecard of the group to which they report (Kaplan and Norton, 1996).

<u>Lag Measure (Performance Measures)</u>: are measures focusing on results at the end of a time period. They normally characterize historical performance and are easy to identify and capture (Niven, 2006).

<u>Lagging indicators:</u> measures that represent the consequences of actions previously taken, which it also referred to as lag-indicators. They frequently focus on results time and characterize

historical performance. Employee satisfaction may be considered a lag indicator. A good balanced scorecard must contain a mix of lag and lead indicators (Kaplan and Norton, 1996b).

ALERT: All Africa Leprosy, Tuberculosis, Training and Rehabilitation Center.

1.9 Organization of the Study

The paper was organized in five chapters. The first chapter presents the introductory part consisting of the background of the study, statement of the problem, research questions, objectives of the study, definition of terms, significance of the study, and scope of the study.

The rest of the paper was organized as follows: Chapter two presents empirical and theoretical review of the literature related to the concept of balanced scorecard, its implementation and challenges faced; Chapter three provides research design and methods employed in the analysis; Chapter four contains the data presentation, analysis and interpretation; and Chapter five gives the conclusion and recommendations. A reference of related literature that was referred while writing the paper and appendices are included after chapter five.

CHAPTER TWO

REVIEW OF RELATED LITERATURE

The main purpose of this chapter is to present the theoretical and empirical understanding of the concept of the balanced scorecard. After a short introduction of the concept and its origin, the evolution of the Balanced Scorecard is described and what it takes to implement it. In the final parts of the chapter, the key success factors of implementing the balanced scorecard and the conceptual framework of the study is stated.

2.4 Theoretical Literature Review

2.4.1 Brief Overview of Performance Measurement

Every organization should measure, monitor and analyze its performance. Performance is defined as an accomplishment of a given task measured against preset known standards of accuracy, completeness, cost, and speed (Bierbusse & Siesfeld, 1997). Performance measurement is a complex issue that normally incorporates at least four disciplines: economics, management, accounting and information technology (Tagen, 2004).

Performance measurement can be defined as the process of quantifying the efficiency and effectiveness of action (Neely, Gregory & Platts, 2005). The function of measurement is to develop a method for generating a class of information that will be useful in a wide variety of problems and situations. Performance measurement is a mystery, complex, frustrating, difficult, challenging, and important, abused and misused (Sidrova & Isik, 2010).

The main objective of PPMS is to provide comprehensive and timely information on the performance of business processes. This information can be used to communicate goals and current performance of a business process directly to the process team, to improve resource allocation and process output regarding quantity and quality, to give early warning signals, to make a diagnosis of the weaknesses of a business process, to decide whether corrective actions are needed and to assess the impact of actions taken (Kueng, 1998).

Kaplan and Norton (2001) began by arguing that an organization's measurement system strongly affects the behavior of managers and employees. They went on to say that "traditional financial accounting measures, like return-on-investment, can give misleading signals for continuous improvement and innovation." To counter the tendency to rely too heavily on financial accounting measures, Kaplan and Norton argued that senior executives should establish a scorecard that takes multiple measures into account (Tupa, 2010).

Kaplan and Norton divide measures into four categories of perspective (Sinclair & Zairi, 1995). The four perspectives of the Scorecard approach, thus, represent one of the most prominent inventions of these frameworks which significantly transformed organizational performance measurement systems. Probably the most well-known approach to performance measurement is the Balanced Business Scorecard (BSC), proposed by Kaplan and Norton (2001).

2.4.2 The Origins of the Balanced Scorecard

The Balanced Scorecard was developed by Robert Kaplan, a Harvard University professor, and David Norton, a consultant from the area of Boston. In 1990 they started research in several companies with the aim of exploring new methods of performance measurement. Traditionally, industries had been relying mainly on financial measures to indicate performance. Many criticisms arose about using only financial measures to track organization performance. In their study, Kaplan and Norton argue that financial measures were too one sided and not relevant to many levels in the organization and that reliance only on financial measures may affect the ability of organizations to create value (Niven, 2006).

Kaplan and Norton (1992) argue that focusing exclusively on financial performance measurements worked well in the era of industrialization, but in the era where new competences were emerging, financial measurements are not enough. Niven (2006) indicated some criticism of the excessive use of financial measures:

- The rising importance of intangible assets. Traditional financial measures are not designed to capture the aspects or performances of customers, suppliers, employees, company culture, quality, and opportunities for learning and innovations. Performances of these intangible assets should be measured because they represent the operational drivers for future financial performance.

- **No predictive power for the future.** Even if financial measures are an excellent summary of past achievements, they are not able to show the right path for future activities and events.
- They do not represent cross-functional and team-work activities. A great deal of business value is created by the collective efforts of different functional areas. Financial statements, on the other hand, represent individual achievements of different functional areas summarized in the overall company picture. They are not able to track the various relationships which continuously develop within an organization in different functional areas.
- **Short-term view**. Focusing only on financial measurements may harm long-term success. In contrast to activities which bring results in the long term such as research and development, employee training, or customer relationships, cost reductions may lead to temporally better financial results but threaten future development and the creation of long-term value.
- They do not involve all levels of an organization. To effectively perform their daily activities, all employees need performance information. Financial measures often involve information from all levels summarized in high-level financial statements. Data presented in such a manner may not be very useful because very often they do not reach all the levels of the organization and its employees.

Some practitioners argue that managers can hardly work with multiple measurements of performance. However, Kaplan and Norton (1992) make an analogy with an airplane cockpit. They explain that for the complex task of navigating a plane, the pilot should rely on a number of indicators and instruments to reach the destination safely and efficiently. He needs detailed information about fuel, airspeed, pressure, altitude, destination, and other indicators that summarize the current and predicted environment. Relying only on one instrument could be fatal. The same can be said for organizations. Managers should recognize the need to track performance in several areas. The Balanced Scorecard should therefore provide answers to four basic questions that look at the business from four important perspectives (Kaplan and Norton, 1992):

- How do customers see the business? Customer Perspective
- What is it important to excel in? Internal Process Perspective

- Can the business continue to improve ability and create value? Learning and Growth Perspective
- How do shareholders see the business? Financial Perspective.

2.4.3 The Concept of Balanced Scorecard

Balanced scorecard is a management tool that enables an organization to translate its vision and strategy into a tangible set of performance measures. However, it is more than a measuring device (Kaplan and Norton 1996).

It is a strategic planning and management system that is used in business and industry, government, and nonprofits organizations to align business activities to the vision and strategy of the organization improve internal and external communications, and monitor organization performance against strategic goals. The balanced scorecard was developed by Robert Kaplan, an accounting professor at Harvard University, and David Norton, a consultant in Boston. In 1900s, Kaplan and Norton led a research study of a dozen companies exploring new methods of performance measurement. The drive for the study was a growing belief that financial measures of performance were ineffective for the modern business enterprise. They discussed a number of possible alternatives but settled on the idea of a scorecard featuring performance measures capturing activities from throughout the organization customer issues, internal business processes, employee activities, and of course shareholder concern (BSC institute 2015).

The key claims success by Kalpan and Norton (1996) are that the BSC demonstrates link to strategy, cause and effect relationships and balance between BSC metrics. They also stated that the BSC is used as a communication tool, a measurement tool or a strategic management tool (Kalpan and Norton 2001). A key conclusion of the research was that in order to be successful, a company has to address long term issues and not simply rely on producing good short term results. Kalpan and Norton argue that financial measures were too one sided and not relevant to many levels in the organization and that reliance only on financial measures may affect the ability of organizations to create value (Niven, 2006).

Moreover, Kalpan and Norton (1999) argue that focusing exclusively on financial performance measurements worked well in the area of industrialization, but in the era where new trends are emerging, financial measurements are not enough. Kalpan and Norton (1992) made an analog

with an airplane cockpit. They explain that for the complex task of navigating a pilot, the pilot should rely on a number of indicators and instruments to reach the destination safely, and efficiently. He needs detailed information about fuel, air speed, pressure, altitude, destination, and other indicators that summarize the current and predicted environments. Relying only on one instrument could be fatal. The same can be said for organization. Managers should recognize the need to track performance in several areas.

Thus, the Balanced Scorecard (BSC) is specific tool that allows managers to define and implement a set of financial and nonfinancial indicators in a balanced way to assess an organization's performance from four viewpoints. The balanced scorecard should therefore provide answer to four basic questions that look at the business from four important perspectives (Kaplan and Norton, 1992). These are customer perspective, internal perspective, financial perspective and learning. The details are discussed below.

2.4.4 Balanced Scorecard Perspectives

With the BSC framework, four categories of measures are identified in order to achieve balance between the financial and non-financial, between internal and external and between current performance and future performance (Kaplan et al., 1992).

Kaplan and Norton do not disregard the traditional need for financial data. Timely and accurate funding data will always be a priority, and managers will do whatever necessary to provide it. In fact, often there is more than enough handling and processing of financial data. With the implementation of a corporate database, it is hoped that more of the processing can be centralized an automated. However, the point is that the current emphasis on financials leads to the "unbalanced" situation with regard to other perspectives. Balanced Scorecard concept offers to look at an organization's strategy from the following points of view (perspectives).

2.4.4.1Financial Perspective

How Do We Look to Share Holders?

Financial measures indicate whether the company's strategy implementation and execution are contributing to the bottom line improvement (Kaplan and Norton, 1992, p.77). According to Paul R.Niven (2006), the Financial Perspective focuses on measures which have the goal of enhancing shareholder value. Possible performance measures under this perspective are derived

from the objectives of revenue growth and productivity which include profitability, cost/unit, revenue growth, asset utilization etc.

Niven (2006), also states that focusing resources, energy, and capabilities on customer satisfaction, quality, knowledge, and other factors in the rest of the perspectives without incorporating indicators showing the financial returns of an organization may produce little added value.

2.4.4.2 Customer Perspective

How Do Customers See Us?

The customer dimension of the Scorecard contains several core or general measures which include high degree of satisfaction of clients, customer retention, attraction (acquisition) of new clients and market share etc.

According to Kaplan and Norton (1996a), the objective of an organization with this perspective is to identify the customer and market segments in which the organization will compete and, accordingly, the measures to track related performances. This enables an organization to identify and focus on factors that are really important in meeting customers 'demands.

Niven (2006) also notes that to achieve positive financial results, organizations need to create and deliver products and services which customers perceive as adding value to them. He remarks that the measures in the customer perspective should answer three basic questions: What are our target groups of customers? What do they expect or demand from us? What would the value proposition for us be in serving them?

According to Kaplan and Norton (2000), the value proposition may be chosen within three differentiators:

- Operational excellence focus on low price and convenience;
- Product leadership offer the best product in the market;
- Customer intimacy focus on long-term customer relationship through a deep knowledge of their needs.

Niven (2006), also states that the most common measures for this perspective include: customer satisfaction, customer loyalty, and market share.

2.4.4.3Internal Business Process perspective

What Must We Excel at?

The internal business process perspective of the BSC mainly emphasizes that organizations must control important working conditions or inner process that may create value to customers and shareholders. For an organization to be effective in performance, the internal working process and systems should be as excellent as possible in assisting operational units provide values to attract and retain clients of the market.

Kaplan and Norton (1992), presume that great customer performance is the result of processes, decisions, and actions which managers need to focus on in order to satisfy customers' needs. This perspective measures the business processes that have the greatest impact on customer satisfaction. Niven(2006) highlighted that this perspective measures an organization's performance with respect to speed (on-time delivery, process cycle time, customer response time etc), quality (continual improvement, rework, repair and scrap, process capability etc), Measures of cost (costs of waste, cost per transaction etc) and other measures (floor space utilization, forecasting and planning accuracy etc).

2.4.4.4Learning and growth perspective

Can We Continue to improve and Create Value?

Internal measure for the scorecard should stem from the business processes that have the greatest impact on customer satisfaction factors that affect cycle time, quality, employee skills and productivity (Kaplan and Norton, 1992, p.75). This perspective answers the fundamental question to achieve our targets and accomplish core activities, how must we learn, communicate & work together?' and it is the foundation up on which the balanced scorecard is built.

Niven (2006), notes that measures of the Learning and Growth perspective are the enablers of the other perspectives. It also emphasizes that knowledge, employee skills and satisfaction, the availability of information and adequate tools are frequently the source of growth and therefore the most common measures of this perspective.

Organizational learning and growth come from three principal sources: people, systems, and organizational procedures...businesses will have to invest in reskilling employees, enhancing information technology and systems, and aligning organizational procedures and routines (Kaplan and Norton, 1996, pp.28-29).

To succeed financially, Financial Initiative Objective how should we appear Measure Target to our owners? Vision Internal Business Processes Customer Objective Measure Target Initiative Objective Measure Target Initiative Mission Strategy To achieve our To satisfy our vision, how customers, at should we appear Learning & Growth what business to our customers? Objective Measure Target Initiative processes must we excel? To achieve our vision, how will we sustain our ability to learn and improve?

Figure 1: Balanced Scorecard Framework/The Four Perspectives of BSC/

Source: Robert S. Kaplan and David P. Norton (1996)

2.4.5 The Steps of Building and Implementing Balanced Scorecard

There are nine steps to build and implement BSC successfully in an organization as designed by Kaplan and Norton (1996). The first six steps are the building (designing) steps and the last three steps are the implementation stages of BSC.

Organizational Assessment: as the first step of building balanced scorecard process is the process of conducting organizational assessment of its vision, mission and SWOT analysis so as to determine how ready the organization for embarking balanced scorecard journey.

Developing Strategy: Strategy represents the broad priorities adopted by the organization in recognition of its operating environment and pursuit of its mission. All performance objectives and measures should align with organizational strategy.

Strategic Objectives: Objectives are concise statement of what the organization must do well in each of the four perspectives in order to execute its unique strategy objectives are organization's continuous improvement activities, which should link to organizational strategic themes, perspectives and strategic results.

Strategic Maps: A balanced scorecard strategy map is generic architecture for describing a strategy. The strategy map describes the process for transforming intangible assets in to tangible customer and outcomes. It provides executive with a framework for describing and managing strategy in a knowledge economy. A strategy map helps organizations to see their strategies in a cohesive, integrated and systematic way. The strategy specifies general directions and priorities. Translating the strategy in to action is the next step. Strategy maps provide the tools to translate general strategic statements into specific hypothesis, objectives, measures and targets.

Performance Measures: The balanced scorecard translates an organization's mission and strategy in to a comprehensive set of performance measures that provides the framework for strategic measurement and management system. The balanced scorecard measures organizational performance across the four perspectives: financial, customers, internal business and learning and growth.

Strategic Initiatives: are the specific programs, activities, projects, or actions you will embark on to help ensuring you meet or exceed your performance targets. Initiatives are designed to close the gap between current performance and that embodied in the stretch targets established .The targets is your mind for the performance measure, and to get there, you need to determine what investments and in initiatives are necessary to guarantee a positive outcome.

Software and Automation: Software and automation involves automating the balanced scorecard system, and consists of analyzing software options and user requirements to make the most cost-effective software choice for today and to meet enterprise performance information requirements in the future. Automating your balanced Scorecard provides a number of benefits and maximizes its use as measurement system, strategic management system, and communication tool. The advanced analytics and

decision support provided by even the simplest Scorecard software allow organizations to perform intricate evaluations of performance and critically examine the relationships to among their performance measures. Automation also supports true organization-wide deployment of the tool.

Cascading: the scorecard means driving it down into the organization and giving all employees the opportunity to demonstrate how their day-to-day activities contribute to the company's strategy. Strategic alignment of a business unit must take place in multiple directions. Cascading creates a line of sigh from the employee on the shop floor back to the executive boardroom.

Evaluation: Evaluation is the last step of balanced scorecard implementation which concerns with the ability to determine progress made toward the goals or targets set. The assessment ability is monitoring functions that simply tracks activities (and their results) it may be as simple as a lot to do list or as complicated as plan of action with milestones.

2.4.6 Success Factors of Balanced Scorecard Implementation

Regardless of all the popular benefits that are promised by the balanced scorecard, empirical studies, such as Pujas(2010), show that the following are key success factors of BSC implementation.

- ➤ Balanced Scorecard Concept Clarity: refers to the failure to understand what exactly the Balanced Scorecard is and what its implementation involves.
- **Executive Sponsorship**: represents the support of the top management of the organization during the development and implementation of the system.
- ➤ <u>BSC Education and Training</u>: like any system, the implementation of the balanced scorecard, requires the creation of sufficient awareness which obviously requires conducting continuous training and education.
- Adequate IT Support: According to Niven (2006), the problem of gathering and entering data into the Balanced Scorecard is emphasized during the implementation process. To overcome such problem and smooth the implementation process, the system should be supported by an appropriate IT system.

- ➤ Effective Project Team: To create a Balanced Scorecard that is capable of implementing the company strategy, linking individuals, creating new behavior and enhancing communication, a team of people is needed. Many ambitious initiatives have failed just because they were led by ineffective teams.
- ➤ <u>Participation:</u> According to Pujas (2010), during the implementation process, if the importance of employee involvement is not understood, the organization may miss the opportunity to benefit from the employees' knowledge that is directly related to the areas in which they exert influence.
- Adequate Key Performance Indicators (KPIs): refers to the measurement of the strategic objectives of the balances scorecard. The decision about what metrics to incorporate in the balanced scorecard is perceived as one of the most difficult parts of the initiative (Niven, 2006). Therefore, due consideration should be made by the implementer while designing the scorecard.
- Planning and Communication: similar to any system, the implementation of the balanced scorecard system requires a precise development plan to guide the selected team during the BSC journey. Without a formal plan showing the implementation path in advance, there is the risk of confusion.

Finally, the researcher presupposes that if such success factors are well addressed in the stated hospital, the overall implementation of the balanced scorecard system, as their performance measurement and strategic management system would be successful.

2.2 Empirical Literature Review

2.2.1 Brief Overview of BSC in Africa

Over the past decades, many African countries have reformed their public sector to improve efficiency, accountability and good governance as part of their efforts to improve service delivery & performance. Virtually every country in the region (Africa) has undergone some kind of civil service reform program (Van, W., 2001, p.84).

Currently most organizations in Africa have adopted BSC for their organizational performance measurement and management system to cope with the firm global and local competition. Use of the Balanced Scorecard is growing rapidly on the continent of Africa (BSI Africa, 2015).

However, there are contradicting arguments about the effectiveness of BSC implementation with African context. Some argue that as this initiative tool is originated in a developed country in milieu of the advanced N. American & Western World business settings, it doesn't fit African existing culture.

According to Kamwachale (2011), the socio-cultural disparities between the African and the Western settings is quite wide in that the African framework is humanist, inclusive, socialist, community based in nature and stakeholder oriented, whereas the Western framework is exclusive and shareholder centered. He also argues that, because the current BSC model is developed to fit the Western socio-cultural framework which is based on maximization of shareholders' wealth, it contradicts the African inclusive and stakeholder oriented governance. Thus, he stresses, it is not ideal and helpful to an organization operating in African environment; to implement the BSC effectively in African organizations, it should incorporate suppliers, government, competitors, community and environmental concerns.

However, others argue that failure in implementation of BSC in some African organizations is due to lack of government and management commitment in addressing the basic requirements of BSC right from project initiation and throughout the implementation process in a continuous & consistent manner. Yet there are sufficient evidences (eg. Kaplan, 2010, Nivev, 2006) that BSC promotes significant improvements and enhances efficient resource allocation and utilization, improved work culture and system, enhanced customer satisfaction, and in general greater organizational performance if it is properly implemented and gets on time support, commitment and continuous monitoring from government and management at all levels regardless of level of advancement or cultural aspects of a nation or a continent.

2.2.2 Brief Background of BSC in Ethiopia

Changes in the socioeconomic and political landscape in Ethiopia have resulted in changes in management of public institutions. The quests for efficiency, accountability, and transparency have forced institutions to adopt organizational strategies and management structures that are popular in business organizations. To this effect, Ethiopia has introduced a range of approaches that broadly fall into the category of "New Public Management". The main successive reform tools include Business Process Reengineering (BPR), BSC and the Citizen Charters. Building on

earlier efforts to establish strategic planning norms and an employee result-oriented performance appraisal system, the federal government introduced the BSC in 2010 (Menwuyelet Fentie, 2014).

The BSC is being used as a planning, change management, and communication tool, and is also considered as an opportunity to embed strategic management principles into the plan. The BSC incorporates the higher-level goals identified by the country's Growth and Transformation Plan (GTP) and tries to operationalize these goals. According to Menwuyelet (2014), three reasons are cited as rationales to implement BSC in Ethiopian Federal civil service organizations, namely:

- To have better performance evaluation system in civil service organizations
- To put in practice the Federal Civil Servants" proclamation no. 515/99 (proclamation for administration of Ethiopian Federal Civil Service employees)
- To focus on strategies and link the day to day activities to strategies with well-balanced and multi perspective performance evaluation system.

A ministry or agency provides a highest expression of tangible and identifiable results that the government expects, below the Ministry or Agency level, the results are further cascaded down to process teams and individual members of staff. BSC set targets for output and interventions for which measures or indicators are set. As of 2012, the civil service's "Individual Employee Performance Management and Measurement" directive was incorporated into the BSC. The management of individual performance is now being addressed through a system of individual performance agreements that are linked to team and organizational targets derived from the four dimensions of the BSC. These include: i) finance, ii) learning and growth; iii) client/customer satisfaction, and iv) processes (time volume, frequency). The first set of annual performance agreements for individuals were signed in July 2012. Currently 60 percent of the score is allocated towards achieving the agreed upon specific tasks and goals, while 40 percent is dedicated to attitude or behavioral aspects (this can also be seen as a reaction to earlier criticism of assessments introduced by the BPR that were seen as overemphasizing "attitudes"). Thus, the BSC seeks to create direct links between employees" performance and the strategy of an organization towards its clients/citizens, budget, internal processes, and learning and Journal of Public Administration, Finance and Law growth. Both employee and organization performance reviews are carried out twice a year, in December and June.

2.2.3 Significance of BSC in the Health Sector

The Ethiopian Federal Ministry of Health (FMOH) has made great strides in creating a culture of planning. From strategic to operational planning, members of the health sector at large appreciate the value of planning their work and working their plans. In 2007, the FMOH embarked on a journey to transform its strategic planning and management processes by adopting the Balanced Scorecard (BSC) as their framework. Extensive work has been done at the leadership level to inculcate the sector with the concepts of balanced scorecard and to link day-to-day activity with the organization's long term direction. However, an assessment of the current state of BSC implementation within the sector has revealed that, in order to evolve the FMOH BSC into a truly strategic system and for the Sector to full realize the benefits of a best practice BSC system, significant changes in both the FMOH BSC structure and leadership action must be carried out (Rahel, 2010).

Piloting the balanced scorecard to a division of the Health Sector was perceived to lower the risk of implementation and provide proof of concept to the Ministry Leadership. FMOH decided to send three additional Health Sector employees from the Capacity Building section to Balanced Scorecard Institute training courses in 2006 in preparation for piloting the process to the Federal Ministry of Health (FMOH). For almost two and a half years, Minister Dr. Tedros and his three colleagues led efforts to develop a strategic balanced scorecard for Federal Ministry of Health (FMOH), a division of the Health Sector, with varying levels of success. Realizing their initial FMOH scorecard had too many strategic themes and strategic objectives, FMOH leveraged its participation in the Bill & Melinda Gates-funded Ministerial Leadership Initiative for Global Health to engage the Balanced Scorecard Institute to facilitate the process for the FMOH pilot and provide additional training. From August 2009 to September 2010, with assistance of the Institute, the initial FMOH scorecard was simplified, refined, and recalibrated (Balanced Scorecard Institute, 2013).

After observing the impressive results that were being achieved by the recalibrated FMOH scorecard, Ministry Leadership subsequently decided to adopt an overall Health Sector scorecard. With additional project funding provided by the Bill and Melinda Gates Foundation in 2011-2013, the Ministry of Health began cascading the strategic scorecard throughout all levels of the Health Sector. The cascade work performed in 2011-2013, and the break-through

improvements that the Ethiopian Health Sector achieved as a result of improved strategic direction and alignment using the institute way. In fact, it has been such a success that the former Prime Minister of Ethiopia mandated that all Ministries in Ethiopia adopt the balanced scorecard as a strategic planning and performance management methodology (ibid).

2.3 Conceptual Framework

Balanced Scorecard is a performance measurement and strategic management tool that provides a balance between certain relatively opposing forces of strategy: short term and long term objectives; Internal and external influences; Leading and lagging indicators and measures; Financial and non-financial goals; It also aligns strategic goals with objectives, targets, and metrics and it cascades corporate strategy to all levels of the organization.

Taking these and other detailed benefits into consideration, the government has initiated the adaption and implementation of BSC in all for-profit and not-for-profit public organizations. As a result, all public financial institutions also engaged themselves in the development and implementation of BSC. Currently, ALERT hospital has finalized developing the BSC as their performance measurement system.

Though the organization established an ad hoc team to study and develop the system and adapt it to their context, the implementation process is apparently turns out to be full of challenges. From the preliminary informal observation of the student researcher, the major repeatedly shown challenges are limited awareness about the BSC, limited senior management commitment, problem of keeping the scorecard at the top, and the problem of introducing the balanced scorecard merely for compensation.

However, these challenges are not systematically sorted out and the effect and depth of each of these challenges on the implementation were not structurally studied. As a result, it is not possible to work them out and deal with them accordingly for the smooth implementation of the system.

Apart from these, the student researcher believes that the theoretical concepts of the balanced scorecard were not followed in the organization and this might be the other challenge that makes the organization unable to fully garner the benefits of adopting the system.

The student researcher believes that the implementation practices of BSC are not well studied in the nation. But knowing and closely studying the implementation practices and the obvious challenges faced will initiate decision makers to immediately address the challenges. It also helps others similar organizations take lessons for effective implementation of the same without making similar mistakes made by the already implementing organizations.

Regardless of all the popular benefits that are promised by the balanced scorecard, empirical studies, such as Pujas (2010), show that the following are key success factors of BSC implementation.

- **Balanced Scorecard Concept Clarity**: refers to the failure to understand what exactly the Balanced Scorecard is and what its implementation involves. The researcher tried to measure this challenge by designing four relevant questions that include the meaning and feature of the balanced scorecard and its implementation steps.
- **Executive Sponsorship**: represents the support of the top management of the organization during the development and implementation of the system. Five questions about the commitment and level of support of the top management were designed to measure the status of the stated challenge.
- Lack of BSC Education and Training: like any system, the implementation of the balanced scorecard, requires the creation of sufficient awareness which obviously requires conducting continuous training and education. The study measured the status of this challenge by designing five relevant questions and analyzing the result accordingly.
- Inadequate IT Support: According to Niven (2006), the problem of gathering and entering data into the Balanced Scorecard is emphasized during the implementation process. To overcome such problem and smoothen the implementation process, the system should be supported by an appropriate IT system. The existence and usage of the same has been measured by designing three relevant questions and analyzing the responses gathered.
- **Effective Project Team**: To create a Balanced Scorecard that is capable of implementing the company strategy, linking individuals, creating new behavior and enhancing communication,

a team of people is needed. Many ambitious initiatives have failed just because they were led by ineffective teams.

- Lack of Participation: According to Pujas (2010), during the implementation process, if the importance of employee involvement is not understood, the organization may miss the opportunity to benefit from the employees' knowledge that is directly related to the areas in which they exert influence. The level of participation has been measured by designing four relevant questions.
- Adequate Key Performance Indicators (KPIs): refers to the measurement of the strategic objectives of the balances scorecard. The decision about what metrics to incorporate in the balanced scorecard is perceived as one of the most difficult parts of the initiative (Niven, 2006). Therefore, due consideration should be made by the implementer while designing the scorecard. The appropriateness of the KPI's under study is measured by designing four relevant questions in the standard likert scale questionnaire.
- Planning and Communication: similar to any system, the implementation of the balanced scorecard system requires a precise development plan to guide the selected team during the BSC journey. Without a formal plan showing the implementation path in advance, there is the risk of confusion. To measure the same five relevant questions with five scale likert scale were designed and analyzed.

Finally, the researcher presupposes that if such success factors are well addressed in the ALERT hospital, the overall implementation of the balanced scorecard system, as their performance measurement and strategic management system would be successful.

Figure 2. 3: Conceptual Framework

Key Success factors of BSC implementation

- **▶** BSC Education and Training
- Executive Sponsorship
- → Adequate IT Support
- **▶** Effective Project Team
- Organizational Participation
- → Adequate Key Performance Indicators (KPIs)
- Planning and Communication

Source: Self-Extracted

SCORE CARD
IMPLEMEANTATION AS
PERFORMANCE
MANAGEMENT SYSTEM

CHAPTER THREE

RESEARCH DESIGN and METHODOLOGY

This chapter provides an outline of the research methodology that was used in the study. The plan of the research and the detailed tactics that assisted to answer the research questions and arrive at conclusions are presented.

3.11 Research Methodology

Best and Kahn(2006) says research can be qualitative, if it describes events and persons scientifically without the use of numerical data while quantitative research consist of research in which data can be analyzed in terms of numbers. Mixed approach is an approach, which combines both qualitative and quantitative ones. Thus, the study applied mixed approach that is quantitative and qualitative approaches as research methodology. Mixed approach helps to make investigation the effectiveness of BSC implementation on performances and to gain a better understanding by obtaining data from different sources, such as interviews and questionnaire which helps to helpful to describe and interpret the trend of events that exist and consequently helps to increase the validity, reliability of the findings and comforts data analysis. It was also aimed to better meet the objectives of the study set out under the first chapter.

3.12 Research Design

According to (Heron, 1998 as cited In Newman, 2013) a research design refers to a plan for collecting and utilizing data so that desired information can be obtained with sufficient precision. A research design is a plan according to which a researcher obtains research participants and collects information from them. This study applied descriptive research design. Descriptive research studies are those studies which are concerned with describing the characteristics of a particular individual, or of a group (Kothari, 1990).

The primary purpose of this study is to describe BSC as a performance management system in ALERT in light of the theoretical framework and hence it can be said that it has a descriptive nature. The major purpose of descriptive research is to describe characteristics of a certain phenomenon. This research also aims to describe the BSC as a performance management system in ALERT. Descriptive research designs describe the characteristics of objects, people, or organizations (Zikmund, Babin, Carr & Griffin, 2012:15). Descriptive research paints a picture

of the specific details of a situation, social setting, or relationship. By giving answers to who, what, when, where, and how questions.

3.13 Data Sources

In this study both primary and secondary data are collected. Relevant information to the study will be obtained both from primary and secondary sources. Primary data are those which are collected afresh and for the first time, and happen to be original in character (Kohtati, 2004). Primary data were collected from top level managers, workers and directorate of policy and planning with specific team of BSC, which are the results of questionnaires and interview.

Secondary sources are those which are made available or have been collected for other research purposes. It refers to data that are already available i.e. data which have already been collected and analyzed by someone else (Kohtari, 2004). The secondary data were collected from different published materials like report and manuals of the MOH, websites, research findings, policy documents and other relevant materials were reviewed.

3.14 Data Collection Techniques

In order to obtain relevant and adequate information the researcher used questionnaire and interview as instruments of data collection. Interview is opted for since its potential as a data collection method can be best exploited when used for the exploration of more complex and subtle phenomena (Denscombe, 2007). This plus its ability to enable the researcher gain insights into the respondents' opinions, feelings, emotions and experiences makes the interview method appropriate in this case.

Among forms of interview semi-structured interviewing method was used in this case study. Semi-structured interview involve series of open-ended questions based on the topic areas the researcher wants to cover, it gives chance to move the conversation in any direction of interest that may come and also used to ask questions which were not included in the structured interview in case new questions raised as ideas emerge through the process. The interview was conducted in the office of the respondents and followed the framework provided under the interview guide who is attached under Appendix no II.

The questionnaires were addressed in close ended form and Lykert type scale of measurement was used to determine the level of agreement or disagreement represented by numerical values ranging from one to five. Five scale formats namely strongly agree, agree, moderately agree, disagree, and strongly disagree. Close ended format is chosen as it is easier and quicker for respondents to answer, easier to compare the answers of different respondents easier to code and statistically analyze. The questionnaires distributed to the management and other employees are different as both have different roles.

3.15 Study area and population

The pools of population for this study were non-management, management and former BSC team members of ALERT. Only employees who have Diploma and above and worked for more than a year in the organization were included in the pool as recently recruited employees may not have deep knowledge of the benefits and root causes of challenges of BSC implementation in the organization as well as won't be able to see at least one cycle of the implementation of BSC. The number of employees working at ALERT who had Diploma and above and a year or more experience as of May, 2020 was 750 (of whom 90 management and 591 non-management members).

3.16 Sample and Sampling Techniques

Regarding sampling technique, Non-probability sampling, specifically purposive sampling technique is used by purposely selecting individuals and groups which the researcher believed has better access to the development and implementation of the balanced scorecard system. The study adopts purposive or non-probability sampling method. This sampling method involves purposive or deliberate selection of particular units of the universe for constituting a sample, which represents the universe.

This sampling technique is appropriate for selects cases that are especially informative for the specific study, and it is also relevant for conducting case study analysis to find important individuals and/or groups that are important for the study. As said by Maxwell (1999), purposive or judgmental sampling is a non-probability sampling technique that uses judgment and deliberate effort to pick individuals who meet specific criteria of the study.

The sample size distribution was given in the manner based on the total employees of the organization under study and the time taken by the organization in adopting the balanced scorecard system. The respondent's composition was from the management (Directors, and Team Managers/Leaders) and purposively selected non-management professional level employees. The desired sample size is 260.

This is calculated using the following sample size determination formula which was developed by Yamane Taro (1967). The reason for using this formula is that it is the most simplified and widely applied in determining sample size in such research works. A 95% confidence level and e = 0.5 precision are assumed for the Equation.

$$n = N/1 + N(e)2$$
 Where

N = Target population

n = Sample size

e = Level of precision=0.05

Given confidence level of 95% and precision rate of ± 5 percent

$$n = 750 / (1 + 750(0.05)2) = 260$$

3.17 Data Analysis Methods

Appropriate statistical analysis such as frequencies and percentiles were used. The data analysis of the study were summarized using frequencies and percentages as well as mean and standard deviation for all variables including age, sex, working experience; years on current job/position and educational level were analyzed tried to be discussed to get meaningful information about existing practices and challenges of balanced scorecard implementation.

The data was presented using tables after collecting and sorting all relevant primary data using the data collection tools, responses were sorted, coded, computed and analyzed and presented in the form of charts, diagrams and tables using Statistical Package for Social Sciences (SPSS) version 20. Besides, the decision rules used in the analysis was average mean less than 3 was considered as low, average mean equal to 3 was considered as medium and average mean greater than 3 was considered as high throughout the study (Best and khan ,1995 8th ed.).

3.18 Reliability

According to Nunnally and Bernstein (1994) stated in Sixholo (2011), Cronbach's alpha is used to test the reliability of the quantitative questionnaire for internal consistence. An alpha value with a lower limit of 0.7 and upper limit of 0.9 was considered acceptable. The reliability test run for the questionnaire of the study showed Cronbach's alpha of 0.87 showing an acceptable internal consistence.

3.19 Validity

Validity is the most critical criterion that indicates the degree to which an instrument measures what it is supposed to measure (Kohtari, 2004). Validity refers to the extent to which the results of the study are accurate. Findings of the study are, thus, presented based on actual results with utmost objectivity of the researcher. Further, to minimize any possible biasing effect and error, Software Package for Social Sciences (SPSS) was applied in analyzing findings of the questionnaires. The researcher also tried to assure the validity of the research instruments in close consultation, comments and recommendations of the advisor.

3.20 Ethical Considerations

The researcher made use of different data collection instruments from different sources. Utmost effort was exerted to acknowledge materials referred and the researcher takes the responsibility to keep confidentiality of respondents 'opinions and unanimity of the rest of the information.

CHAPTER FOUR

DATA ANALYSIS AND INTERPRETATION

This chapter basically provides the data analysis and discussion part of the study. The data collected from respondents was systematically presented and detail analysis of the findings from the questionnaires and interview was conducted and presented with detail interpretation.

4.1 Personal Information of Respondents

The total questionnaire distributed to the ALERT was 260. The total questionnaire filled out and returned back was 225, which is 87% response rate. Summary of demographic characteristics of respondents who filled & returned the questionnaires is tabulated as follows:

Tables-4.1.1 Personal Information of Respondents

No.	Item	Description	Frequency	%
	Gender	Female	107	47.6
1.		Male	118	52.4
		Total	225	100
		18-27	13	5.8
		28-37	118	52.4
2.	Age	38-47	71	31.6
		48-57	21	9.3
		>58	2	.9
		Total	225	100
		Diploma	24	10.7
		Bachelor Degree	116	51.6
3.	Educational Level	Masters	59	26.2
		PhD	26	11.6
		Total	225	100
		1-5	116	51.6
		6-10	53	23.6
		11-15	48	21.3
4.	Year of Experience	16-20	3	1.3
		21-25	2	.9
		>26	3	1.3
		Total	225	100

As can easily be understood from table 4.1.1 above, majority of the respondents (52.4%) were male and the remaining (47.6%) female.

Concerning age of the respondents, most of them (52.4%) fall under the range of 28-37 years, 31.6% between 38-47 years, 9.3% between 48-57 years, 5.8% between 18-27 years, and the remaining .9% aged more than 58 years.

With regard to educational level of respondents, majority of them are first degree holders for 51.6% followed by second degree holders 26.2%, while the remaining were PhD and Diploma holders for 11.6% & 10.7% respectively.

Concerning the level of experience of respondents in the hospital, as indicated in the table, 51.6% of them had worked for 1 to 5 years, 23.6% for 6 to 10 years, 21.3% for 11 to 15 years, 1.3% for 16 to 20 years and 21-25 and the remaining.9% of them had work experience of >26 years in ALERT.

4.2 Analysis of the Variables

To answer the research questions, eight variables that refer to the common challenges of balanced scorecard implementation based on the study of Pujas (2010) which are limited understanding of BSC, lack of executive sponsorship, lack of BSC education and training, inadequate IT support, inadequate project team, organizational participation, inadequate key performance indicators (KPIs) and lack of planning and communication are taken to design the questionnaire and analyse the response of employees and the management of ALERT Hospital. These variables are the major success factors for effective implementation of the balanced scorecard system as performance management tool.

The researcher for each variable has managed to design three to five questions to better represent the variable from different directions. After inserting the raw Likert scale data to SPSS 20, the researcher has analysed and summarized the responses by taking the average of the response results obtained under each variable to arrive at the average response towards the stated variable. Hereunder are the questions under each variable and the analysis of the summarized responses for the questions.

4.2.1 Concept Clarity of the Balanced Scorecard

According to Kaplan and Norton (1992), one of the key factors for the successful implementation of the balanced scorecard is to create clarity of the concept of the balanced

scorecard system weigh before starting implementation of the system. To check whether the concept is absorbed in organization, the sample respondents were asked to express their level of agreement with some basic questions about the concept of the balanced scorecard.

Tables-4.2.1 Concept Clarity of the Balanced Scorecard

Questions		Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	Mean
BSC links short term operational performance with	N	-	103	57	60	5	2.85
long term strategic objectives	%	ı	45.8	25.3	26.7	2.2	
BSC creates ability to translate	N	2	6	44	148	25	3.84
vision into operational strategy	%	.9	2.7	19.6	65.8	11.1	
The ALERT used BSC to set	N	2	8	44	146	25	3.82
business strategies and objectives	%	.9	3.6	19.6	64.9	11.1	
BSC rolls down vision from	N	3	6	44	145	27	3.83
corporate to division, to individual employees	%	1.3	2.7	19.6	64.4	12	

Source: Sources: Survey Result (2020)

Our objectives.distribution and result for implementation of BSC program Av. Mean <3 Low, Av. Mean=3 Medium, Av. Mean> 3 High,

Accordingly, a minimum mean response of 2.85 and the maximum mean response of 3.84 was registered and representing a significant level of agreement. This implies that the hospital under study, the basic concept and awareness about the balanced scorecard not encouragingly created.

4.2.2 Executive Sponsorship

The Balanced Scorecard implementation effort is the crucial part. Suffice it to say that if the company's leader is not aligned with the goals and objectives of the Balanced Scorecard and does not believe in the merits of the tool, all the efforts will be severely compromised. An executive sponsor must provide leadership for the program in both words and deeds According to Niven (2006).

The study tried to measure the opinion of the sample respondents on whether they agreed on the level of executive sponsorship in ALERT.

Tables-4.2.2 Level of Executive Sponsorship

Questions		Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	Mean
Top Management of the ALERT took the initiative	N	1	6	42	151	25	3.86
	%	.4	2.7	18.7	67.1	11.1	
There was full support from the management	N	0	108	57	60	0	2.78
	%	-	48	25.3	26.7	-	
BSC is one of the priorities of	N	2	8	42	148	25	3.83
the ALERT	%	.9	3.6	18.7	65.8	11.1	
Top management periodically	N	0	114	55	56	0	2.74
monitors progress of BSC	%	-	50.7	24.4	24.9	-	
Top Management works closely	N		113	52	60	0	2.76
with the champion	%		50.2	23.1	26.7	-	

To a question presented to them with regard to Top Management of the ALERT took the initiative implementation, the respondents agreed to the management's commitment for the implementation depicting mean score of 3.86. On the other hand, minimum mean score of 2.74 was registered to the question requesting the level of Top management periodically monitors progress of BSC. This implies that the executives of the ALERT under study have not sufficiently supported the implementation process by taking the BSC as one of the priorities of the hospital. However, as they have loads of other responsibilities, there is a tendency of overlooking closely working with the BSC champion.

4.2.3 Balanced scorecard education and training

According to Niven(2007), organizations conduct awareness sessions during the time the Scorecard is declared as a measurement system featuring financial and nonfinancial measures, but little information is offered about the many refinements and difficulties of the model. Often the deceptive simplicity of the Scorecard makes people susceptible to the false notion that indepth training is not required.

Organizations, therefore, should take the necessary time at the beginning of the implementation to develop a comprehensive Scorecard curriculum that includes background on the concept, objectives in implementing the BSC, typical problems, success stories, and practical implementation details. To measure the level of engagement of this organization in education and training, the following five questions were designed in the questionnaire distributed.

Tables-4.2.3 Balanced scorecard education and training

Ouestions		Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	Mean
The Hospital has provided me	N	1	6	42	151	25	2.76
training about the meaning of BSC	%	.4	2.7	18.7	18.7	21.8	
I have been informed about	N	0	108	57	60	0	3.84
corporate objectives of the ALERT.	%	-	48	25.3	26.7	-	
The ALERT informed all employees	N	2	8	42	148	25	2.77
to clearly understand corporate level objectives	%	.9	3.6	18.7	65.8	11.1	
I understood alignment of my unit's	N	0	114	55	56	0	2.79
objectives with the corporate level objectives	%	-	50.7	24.4	24.4	1	
The ALERT provided me training to	N	0	113	52	60	0	3.84
design my personal scorecard	%	-	50.2	23.1	26.7	-	

The level of education and training with regard to BSC is insignificant, with a minimum mean score of 2.76 for a question presented to respondents if their hospital has provided them training that equipped them sufficient knowledge about the meaning of BSC. On the other hand, the maximum mean score of 3.84 is recorded for the question presented to the respondents if their respective hospital has provided them training to design their personal scorecard. This implies that totally, the level of training and education conducted by ALERT is not to the level that can create sufficient knowledge about the balanced scorecard and did not guide them in how to design their own scorecard.

IT Support

The Balanced Scorecard automating provides a number of benefits and maximizes its use as a measurement system, strategic management system, and communication tool. The advanced analytics and decision support provided by even the simplest scorecard software allow organizations to perform intricate evaluations of performance and critically examine the relationships among their performance measures. Automation also supports true organization wide deployment of the tool (Niven, 2006). To evaluate the opinion of the respondents with regard to the level of automation of the balanced scorecard in their respective hospital, three standard questions were included in the questionnaire on which the analysis hereunder is conducted:

Tables-4.2.4 BSC Automation

Questions		Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	Mean
The ALERT's BSC is fully automated	N	32	185	8	0	0	1.89
	%	14.2	82.2	3.6	-	-	
BSC is supported by IT in	N	9	194	19	3	0	2.1
collecting, analyzing, reporting and distributing relevant data	%	4	86.2	8.4	1.3	-	
An appropriate IT system is	N	4	214	6	1	0	2.01
designed to help employees to collect data	%	1.8	95.1	2.7	.4	-	

Regarding the questions presented to the respondents if their respective hospital's balanced scorecard system is fully automated, the respondents significantly disagreed with mean score of 1.89 and with regard to the question presented to them if their hospital use other IT systems in the process of collecting, analysing the data to easily make use of the hospital's balanced scorecard system, still they are significantly disagreed to it. As shown from the table above, when the response of the above listed questions is summarized, the result shows that there is not significant deviation on the consensus that there is not any customized IT system designed or acquired to assist the overall implementation of the balanced scorecard. This implies that there is no any customized IT system being used by these hospitals in order to fully take hold of all the benefits that would be provided by the balanced scorecard system.

4.2.4 Competent Project Team

The structure of a central office for strategy implementation may seem to risk reinforcing top-down decision making and inhibiting local initiative, but it does just the opposite. A unit/team with responsibility for the implementation of new systems such as the balanced scorecard becomes a convenient focal point for ideas that percolate up through the organization Kaplan and Norton, (2005).

Performing strategy and implementing a new system usually involves making changes that only a CEO can empower, and the office which is responsible for the implementation of BSC will be most effective when it has direct access to the CEO. The consciousness of the existence of this establishment and its duties are assessed by posing the below three questions.

Table 4.2.5 Effectiveness of Dedicated Project Team

Quantiana		Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	Mean
Questions		Disagree				Agree	
Establishing a devoted process	N	10	32	69	96	16	3.70
ensures success of BSC							
implementation	%	4.4	14.2	30.7	42.7	7.1	
In the ALERT, there is specific	N	5	22	78	80	40	3.56
person to oversee BSC	%	2.2	9.8	34.7	35.6	17.8	
implementation							
The process which is	N	14	40	103	63	5	3.02
responsible to oversee BSC is		6.2	17.8	45.8	28	2.2	
effective		- · · -	1,.0				

Source: Sources: Survey Result (2020)

From the responses, we can observe that the respondents agreed to the concept of the usefulness of establishing an independent unit with mean score of 3.70. In addition, if they have awareness about the existence of this establishment in their own organization, they have loosely agreed with score of 3.56 this implies that though they have limited awareness about this independent office, still strong awareness creation should be made. On the other hand, the respondents have doubts on the effectiveness of the process/department that is established to oversee the implementation of the balanced scorecard system, depicting it with the mean score of 3.02. This implies that the establishment and functionality of these bodies is not to the level that is required for effective support of the BSC implementation.

4.2.5 Organizational Level Participation

According to Kermally,(1997) successfully balanced scorecard system implemented organizations, as their performance measurement and strategic management system have often reported that involving all employees and the management at all levels in the development and implementation of the BSC helps a lot to build a shared interest, and increases each individual's motivation to see the system succeed.

Table 4.2.6: Organizational Level Participation

Questions		Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	Mean
All employees were involved in	N	9	99	69	35	13	2.7
the implementation	%	4	44	30.7	156	5.8	
BSC implementation was	N	16	40	76	78	15	3.16
participatory	%	7.1	17.8	33.8	34.7	6.7	
I was guided by my supervisor	N	7	80	72	51	15	2.94
while I was designing my scorecard	%	3.1	35.6	32	22.7	6.7	
Information on BSC is provided	N	6	46	70	91	12	3.25
timely	%	2.7	20.4	31.1	40.4	5.3	

Source: Sources: Survey Result (2020)

As regards measuring the level of organizational participation in these public hospital, the under listed four questions were raised to the respondents, and according to their response, they significantly disagreed to the idea that all employees of the organization were involved in the implementation process of the balanced scorecard, with mean score of 2.7. Similarly, they have disagreed to the question presented to them if their supervisor's guided them in how to design their personal scorecard, with mean score of 2.94. And with regard to the timely provision of information about the balanced scorecard, the respondents have loosely agreed with the mean score of 3.25.

4.2.6 Key Performance Indicators

Relating to the key performance indicators used in the balanced scorecards at all levels, the clarity and their relevance to measure the performance and their alignment to the respective ALERT's strategies are measured by designing relevant questions.

Table 4.2.7: Key Performance Indicators

Questions		Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	Mean
KPIs are designed based on	N	0	46	90	60	29	3.32
the ALERT's mission and vision	%	-	20.4	40	26.7	12.9	
At all levels there is sufficient	N	17	50	81	49	28	3.10
KPIs to measure objectives	%	7.6	22.2	36	21.8	12.4	
BSC complements the	N	1	17	14	106	87	4.16
financial measures of past performance	%	.4	7.6	6.2	47.1	38.7	
While designing BSC, Data	N	16	52	64	58	35	3.20
collection method and its frequency was set	%	7.1	23.1	28.4	25.8	15.6	

Regarding the arrangement of KPIs with the mission and vision of the organization concerning the adequacy of the respective ALERT's KPIs to measure their strategic objectives, the respondents have a neutral agreement with mean score of 3.32 and 3.20 respectively. On the other hand, for the question raised to the respondents if BSC complements the financial measures of past performance, the respondents have agreed to it with mean score of 4.16. This implies that there is a loose relationship of KPIs with the organization mission and vision and there are inadequate key performance indicators on the respective scorecards.

4.2.7 Planning and Communication

As said by Kaplan and Norton studied, it is managed to be recognized that effective balanced scorecard execution requires an effective planning and communication. This ensures that enterprise level plans are translated in to the plans of the various units and departments; executing strategic initiatives to deliver on the grand plan; and aligning employees' competency development plans, and their personal goals and incentives, with strategic objectives (Kaplan and Norton, 2005).

With reference to the level of planning and communication practices undertaken to implement the balanced scorecard, the following questions were included in the questionnaire distributed to them for which the response of the respondents is analysed hereunder.

Table 4.2.8: Planning and Communication of BSC System

Questions		Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	Mean
The ALERT's Strategy map is	N	18	26	50	109	22	3.40
clear and understandable	%	8	11.6	22.2	48.4	9.8	
BSC is better than previous	N	7	20	29	42	127	4.16
measurement systems	%	3.1	8.9	12.9	18.7	56.4	
BSC is relevant performance	N	9	58	36	71	51	3.43
management tool for the ALERT	%	4	25.8	16	31.6	22.7	
It is the right time for the	N	56	55	47	40	27	2.67
ALERT to implement BSC	%	24.9	24.4	20.9	17.8	12	

As regards the time of implementing the system is right for the respective ALERT's, their response shows that they loosely disagreed about the questions, with mean score 2.67. On the other hand, with regard to the clarity of the respective organization's strategy map and improvement of the system as compared with the previous performance measurement system, they have depicted their level of agreement with mean score of 3.40 and 4.13 respectively. This implies that communication and awareness creation activities of the organization understudy with regard to the benefit and implementation time of the system is weak.

4.3 Analysis of the Interview

As the exertion to collect information with regard to the implementation of the balanced scorecard, concerned managers of the ALERT were interviewed about the overall process they went through, the challenges they have faced and the current status currently at. Accordingly, the responses from the interview were analysed in the following manner.

The ALERT has officially started developing and implementing the balanced scorecard on August 2007. For a year and a half time since then, the organization had established a dedicated project team, composing members from every processes of the organization, in order to assess the benefit of adopting the system to the context of the organization and it has officially launched the implementation process since July 2008.

In advance of implementing the balanced scorecard, an independent office called Office of Strategic Management has been established which is accountable directly to the Director of the ALERT. According to the Strategic Management team manager, the office has been primarily organized by having all the BSC study team members as its permanent staff. The Office of Strategic Management has been primarily dealing with enabling and monitoring the overall implementation of the balanced scorecard system as the organization's performance measurement and strategic management tool.

From the interview with the Manager, the researcher has managed to learn that the organization, after studying the system, has provided an intensive two weeks training to the overall management and has designed a corporate scorecard in balanced scorecard framework. Then the corporate scorecard has been presented to all processes of the organization and then the Executive Management Team has discussed on all corporate objectives and takes each objective on which each process has an influence on.

Accordingly, each process has designed its own scorecard and has been monitoring its performance quarterly since then. With regard to individual level scorecard, according to the manager, the hospital has been faced with difficulties of obtaining key performance indicators that can objectively measure the individual performance but it has been cascading the corporate scorecard down to the team level and measure their performance progress accordingly.

According to the Strategic Management Team Manager, the major challenges faced by the ALERT are lack of clarity of the concept of the balanced scorecard though in depth training was provided, difficulty of finding relevant KPIs, lack of active organizational level participation inability to back the system with relevant IT system.

CHAPTER FIVE

SUMMARY, CONCLUSIONS AND RECOMMENDATIONS

In this chapter, the findings of the study are systematically summarized, relevant conclusions are drawn from the findings and the study has also tried to forward relevant recommendations.

5.1 Summary of Major Findings

The aim of this study was to assess the overall implementation and challenges faced while implementing the balanced scorecard system as a performance measurement, strategic management and communication system in the ALERT hospital and the study also tries to evaluate the challenges faced as compared to the standard literatures of the system.

Standardized questionnaire with 37 questions organized in eight themes in five-point likert scale were prepared and distributed to 260 respondents selected from the organizations under study. While taking sample, non-probability purposive sampling has been adopted to purposively select respondents who are middle level managers (directors and team managers) and senior officers who have the working knowledge of the balanced scorecard system. Of the total questionnaire, it was managed to collect 225 of them, i.e. 87% response rate.

Hence, the analysis was conducted by taking each theme which are supposedly be a common challenge for BSC implementation. Each common challenge/theme has three to five questions that are suitably designed to measure the status of the ALERT's with respect to the specific challenge and subsequent relevant analysis was made.

Along with this, the directors and managers of the dedicated department who are responsible to monitor and oversee the implementation of BSC were interviewed by designing semi-structured interview questions. Based on the interview, the current status, how the system was being implemented, and the way foreword has been gathered and analyzed.

The major findings of the study results from secondary and primary data analysis are summarized as follows:

- As to the standard literature of the system, it has been managed to learn that most of the steps and implementation requirements, except BSC automation, has been implemented by the ALERT.
- > From the questionnaire distributed, it has been managed to summarize the following findings:
 - As regards creating concept clarity, executive sponsorship of ALERT, designing effective key performance indicators, establishing competent project team/dedicated process, the study has shown that the respondents have agreed to the questions presented to evaluate the degree have been practicing with regard to the stated success factors.
 - Instead, the study has shown that with regard to the formal BSC education and training, planning and communication of the overall implementation, organizational level participation in the development and implementation of the system as well as BSC automation (IT support), the respondents have strongly expressed their partial agreement to the efforts being made by their organization.

5.2 Conclusions

On the conducting the study, a literature review of the BSC, its concepts and main characteristics was conducted. Earlier studies were also used to identify the most important barriers to the successful implementation of the BSC.

The study can drive a general conclusion that with regard to how the balanced scorecard is being implemented in the ALERT, the organization have implemented the system almost in a similar manner as compared to the standard way of literature except that the automation of the system was nonexistent in the organization. The study has found out that there are different mechanisms undertaken by the organization with the intention of supporting their balanced scorecard system by IT.

The BSC implementation of key success factors have been observed by this organization though at different level. Though, with regard to creating concept clarity about the system, the organization has made an encouraging effort. Likewise, the sponsorship status of the system by their respective executives and the existence of sufficient KPIs to measure performance are also strong and the establishment of an independent office to oversee the implementation of the

system has been undertaken and its competence and efficiency in taking over the stated task is also adequate.

Instead, there are some critical success factors that should be worked on to further reap the full benefit of the system. These are enhancing BSC education and training, automating the Balanced Scorecard, sustainably ensuring the organizational level participation to further instill the BSC concept and sustain its improved implementation, and strengthen the BSC planning process and its communication to the overall employees.

In general, it appears safe to conclude that though this organization have been implementing the balanced scorecard system in a way that the literature advises, the stated challenges identified in the study have made the organization not to grasp the full benefit the balanced scorecard system.

5.3 Recommendations

According to Niven (2006), no two BSC implementations are completely alike. Further, the same author states that organizations which decided to implement the tool should do so in a way that fits the individual culture, current management processes, and readiness for such a major change initiative. Therefore, the findings, results and recommendations of this thesis cannot be generalized and taken for granted by other companies, researchers or others interested in the topic.

To deed the benefits of implementing the balanced scorecard as a performance management, strategic management and communication tool, the study proposes the following recommendations to be considered by ALERT. Based on the findings of the study and the conclusion drawn the following recommendations have been forwarded so to augment the BSC effort of the All Africa Leprosy, Tuberculosis, Training and Rehabilitation Center/ALERT/.

 ALERT should conduct intensive awareness creation activities to the employees about the meaning, benefit and the techniques of the balanced scorecard system from the balanced perspective of its nature as a strategic management system, performance measurement and strategy communication.

- Newly presented system in an organization, like BSC to be effectively implemented, should involve organizational level participation in its development and implementation.
- The executive management team of ALERT should be set effective implementation plan and monitoring mechanism and its progress should be attentively followed up.
- The not fully formed IT slowed the initiative and caused many frustrations for the system. Therefore, relevant IT-systems, BSC-software, should be implemented to facilitate and exploit the full benefits of the balanced scorecard.
- To manage the overall implementation of the balanced scorecard system should enhance their follow up systems by periodically monitoring implementation gaps and subsequently fill them by organizing awareness creation activities, and formal trainings in order impart the concept and benefit of BSC to all stakeholders.

Finally, the researcher would like to recommend future researchers to further study the effectiveness of the implementation of the BSC system and other tools in the ALERT and in all government organization and also may be better considering other possible alternative for performance management system with regard to the issue for better executions of strategic plan.

Reference

- Ahmed, A.(2012, June). Improving Government Performance: Use of Balanced Scorecard. A paper presented at International conference on business management, New Delhi.
- Adebabay, A. (2011, March). Promoting and strengthening professionalism in the civil service: The Ethiopian case. A paper presented at Capacity Building workshop on Promoting Professionalism in Public Service sector in Africa, Addis Ababa.
- Atkinson, (1997). "Strategy implementation: a role for the Balanced Scorecard?" Management Decision. Vol. 44, No.10, pp.144-160
- Atkinson, H. (2006). "Strategy implementation: a role for the Balanced Scorecard?" Management Decision. Vol. 44, No.10, pp.144-160
- Adebabay, A. (2011, March). Promoting and strengthening professionalism in the civil service: The Ethiopian case. A paper presented at Capacity Building workshop on Promoting Professionalism in Public Service sector in Africa, Addis Ababa.
- Best and Khan (1995) 'Research in education' 8th ed.
- Bible L, Kerr S, Zanini M (2006). 'The balanced scorecard: here and back'. Management Accounting Q., Vo.17, No.4, PP.18-23.
- Bierbusse, P. and Siesfeld, T. (1997), Measures that matter, Journal of Strategic Performance Measurement, Vol. 1, No. 2, pp. 6–11
- Bosilj Vukšić V., Milanović Lj, Škrinjar R., Indihar Štemberger M. (2008), Organizational Performance Measures for Business Process Management: a Performance Measuremet Guideline, IEEE computer society, ISBN 978-0-7695-3114-4.
- Bosilj Vukšić V. i sur. (2006.), Menadžment poslovnih 19. procesa i znanja u hrvatskim poduzećima, Zagreb: Working Paper Series, 06 (05), 1-20
- Brewer, P.C. and Speh, T.W. (2000), "Using the balanced scorecard to measure supply chain performance", Journal of Business Logistics, Vol. 21 No. 1, pp. 75-93.
- David F (2006). Strategic management: concepts & case studies. Upper Saddle River, NJ: Prentice Hall.
- Denscombe, M. (2007). The good Research Guide for Small Scale Social Research Projects. 3rd ed. Mc Graw hill.
- Dragan, P. (2010). Barriers to the Successful Implementation of the Balanced Scorecard the Case of Plava Laguna J.S.C.
- Drucker, P.F. (1993). Strategic Planning Process Background Information, New York, NY Harper&Rowpublisher,availableathttp://www.suno.edu/Chancellor/docs/Section.I

- .Strategic.Planning.Process.and.Background.References.pdf accessed in October 2014.
- Kamwachale J. (2011).Redesigning the Balanced Scorecard Model: An African Perspective. Unpublished, South Africa, University of Pretoria
- Kaplan S., Norton D. (2001), Transforming the Balanced Scorecard from Performance Measurement to Strategic Management, Accounting Horizons, Vol. 15 No. 1, pp. 87-104.
- Kaplan, R.S. and Norton, D.P. (1997). Why Does Business Need A Balanced Scorecard? Journal of Cost Management, vol. 11, no. 3.
- Kaplan, R.S. & Norton, D.P. (2008). Execution Premium: Linking Strategy to Operations for Competitive Advantage, Boston: Harvard Business School Press.
- Kaplan RS, Norton DP, Rugelsjoen B (2010). Managing alliances with the balanced scorecard. Harv. Bus. Rev., 88(1): 114-120
- Kaplan, R.S., & Norton, D.P. (2000). The strategy-focused organization: How balanced scorecard companies thrive in the new business environment. Boston: Harvard Business School Press.
- Kaplan, R.S. and Norton, D.P. (1996). The Balanced Scorecard: Translating Strategy into action, Boston: Harvard Business School Press.
- Kaplan, R. S. and Norton, D. P. (1999). The balanced Scorecard-from strategy till bandling. Goteborg. ISL. Forlag.
- Kaplan, R.S. & D.P. Norton. (1993). Putting the balanced scorecard to work. Harvard Business Review 71(5), 134 140.
- Kaplan, R.S. & Norton, D.P. (1992). The balanced scorecard Measures that drive performance, Harvard Business Review (January-February), pp. 71-79.
- Kohlbacher M., Gruenwald S.(2011), Process orientation: conceptualization and measurement, Performance measurement system design, International Journal of Operations and Production Management, Vol. 25 No. 2, .267-283
- Kohtari, C.R.(2004). Research Methodology; Methods and Techniques. 2nd ed. New Age International publishers.
- Kueng, P. (1998), Supporting BPR through a Process Performance Measurement System, In: Banerjee, P. et al. (Eds.): Business Information Technology Management, Conference Proceedings of BITWorld'98, Har-Anand Publications, New Delhi, pp. 422-434; ISBN: 8124104255.

- Kueng P., Wettstein Th., List, B. (2000), Holistic Process Performance Analysis through a Process Data Warehouse, In Proceedings of the American Conference on Information Systems (AMCIS 2001), Boston, USA.
- Kueng P. (2000), Process performance measurement system: a tool to support process based organizations, Total Quality Management, Vol. 11 No 1., pp. 67-85
- Kueng, P. (1998), Supporting BPR through a Process Performance Measurement System, In: Banerjee, P. et al. (Eds.): Business Information Technology Management, Conference
- Kaplan R.S and Norton, D. P (1992): The Balance Scorecard Measurement that Drive Performance, Harvard Business Review. Vol.1
- Lawrie, G. and Cobbold, I.(2004). "Third-Generation Balanced Scorecard: Evolution of an Effective Strategic Control Tool." International Journal of Productivity and Performance Management, Vol.53 No.7, pp. 611-623.
- Mesfin Taffesse (2009). The Ethiopian Civil Service Reform Program: In Taye Assefa (Ed.): Digest of Ethiopia's National Policies, strategies and programs. FSS. Addis Abab
- MoCS (2013, April). The Civil Service Reform Program in Ethiopia. Ministry of Cevil Service. Addis Ababa
- MoFED, (2012). Annual Progress Report for FY 2010/11. Addis Ababa, Ethiopia.
- Molleman, B. (2007). The Challenge of Implementing the Balanced Scorecard. Available at http://referaat.cs.utwente.nl/conference/6/paper/6800/the-challenge-ofimplementing-the-balanced-scorecard.pdf. Accessed in November, 2014
- Neely A., Gregory M., Platts K. (2005), Performance measurement system design, International Journal of Operations and Production Management, Vol. 25 No. 12., pp. 1228-1263
- Neely A. (2005), The evolution of performance measurement research, International Journal of Operations and Production Management, Vol. 25 No. 12., pp. 1264-1277
- Neely A., Gregory M., Platts K. (2005), Performance measurement system design, International Journal of Operations and Production Management, Vol. 25 No. 12., pp. 1228-1263.
- Niven, P. (2005). Balanced Scorecard Diagnostics: Maintaining Maximum Performance. John Wiley & Sons, New Jersey.
- Niven, P. (2006). Balanced Scorecard Step-by-Step: Maximizing Performance and Maintaining Results (2nd Ed.). John Wil ey & Sons, New York.
- Nørreklit,(2000). The balanced scorecard: What is the score? A rhetorical analysis of the balanced scorecard, Accounting, Organizations and Society, 28 (6), pp. 591-619.

- Poll, R. (2001), "Performance, processes and costs: managing service quality with the balanced scorecard", Library Trends, Vol. 49 No. 4, pp. 709-17.
- Proceedings of BITWorld'98, Har-Anand Publications, New Delhi, pp. 422-434; ISBN: 8124104255.
- Pujas, D. (2010). Barriers to the Successful Implementation of the Balanced Scorecard. Vienna: Modul University Vienna.
- Rigby,(2001):Management Tools and Trends 2009, Bain & Companyhttp://www.bain.com/management_tools/Management_Tools_and_Trends_2009_Global_Results.pdf. (Accessed on 05 May 2019).
- Robbins P. Stephen, (2004). Organizational Behaviour. Delhi: Baba Barkha Nath Printers.
- Ruben, D. (1999). Toward A Balanced Scorecard for Higher Education: Rethinking the College & University Excellence Indicator Framework. New Brunswick, NJ: Transaction Publicatn
- Saunders, M., Lewis, P. & Thornhill, A. (2009). Research Methods for Business Students. 5th ed. Prentice Hall.
- Seokjin, K., Behnam, N. (2008), The dynamics of quality costs in continuous improvement, International Journal of Quality & Reliability Management, Vol. 25 Iss: 8, pp. 842 859
- Sidrova A., Isik O. (2010), Business process research: a cross disciplinary review, Business Process Management Journal, Vol 16 No. 4, pp. 566-597
- Sinclair D., Zairi M. (1995), Effective process management through performance measurement Part I, Business Process Reengineering and Management Journal, Vol. 1. No. 1, pp. 75-88.
- Tagen S. (2004), Performance measurement: from philosophy to practice, International Journal of Productivity and Performance Management, Vol. 53 No. 8, pp. 726-737
- Tupa, J. (2010), Proces Performance Measurement as Part of Business Process Management in Manufacturing Area, Process Management, INTECH, ISBN 978-953-307-085-8
- Van, W. (2001). African Economies and the Politics of Permanent Crisis. Cambdrige, Cambridge University Press
- Yamane, T.(1967). Statistics: An Introductory Analysis, 2nd Ed. New York: Harper and Row
- Zemetis, K. (2009). Ethiopian Federal Ministry of Health Assessment Status Report.(Ethiopia Balanced Scorecard Overview, 2010).
- Zikmund, W.G., Babin, B.J., Carr, J.C. & Griffin, M. Business Research Methods. 8th ed.

APPENDEXTES

APPENDIX-I

Questioner on Balanced Score Card (BSC)

Dear staff of ALERT,

I am a MA student in St. Mary's University interested in collecting your opinions .Thus, your thoughtful and honest responses to this questionnaire are very important. Your responses will be confidential. This questionnaire contains statements about BSC effectiveness on organizational performance in ALERT Hospital. Hence, the researcher kindly requests you to respond each item carefully.

DEMOGRAPHIC PROFILE

Answe	er the following questions by putting "\sqrt{"}" mark on the box or writing on the space provided
1.	Gender: Female Male
	Age: 18-27 28-37 38-47 48-57 58 & above
2.	Education Level
	Diploma Bachelor
	Masters PHD
3.	How long you have been employed in ALERT?
	1-5 years 6-10 years 11-15 years
	16-20 years 21-25 years 26 and above years
4.	Have you ever attend any training related to BSC?
	Yes No I don't know
5.	Have you ever have BSC planning and implementing experience?
	Yes No I don't know

Answer the following questions by putting " $\sqrt{}$ " mark on the box.

Rating: - 5=Strongly Agree, 4= Agree, 3=Neutral, 2=Disagree and 1=Strongly Disagree

No.	Description/Questions			Rating		
		Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
1.	Concept of the Balanced Scorecard (BSC)					
1.1	BSC links short term operational					
	performance with long term strategic					
	objectives					
1.2	BSC creates ability to translate vision into					
	operational strategy					
1.3	The ALERT used BSC to set business					
	strategies and objectives					
1.	Executive Sponsorship					
2.1	BSC rolls down vision from corporate to					
	division, to individual employees					
2.2	Top Management of the ALERT took the					
	initiative					
2.3	There was full support from the					
	management					
2.4	BSC is one of the priorities of the ALERT					
2.5	Top management periodically monitors					
	progress of BSC					
2.6	Top Management works closely with the					
	champion					
2.	Balanced scorecard education and training	_				
3.1	The Hospital has provided me training about					
	the meaning of BSC					
3.2	I have been informed about corporate					
	objectives of the ALERT.					

	mission and vision			
7.2	At all levels there is sufficient KPIs to			
	measure objectives			
7.3	BSC complements the financial measures of			
	past performance			
7.4	While designing BSC, Data collection			
	method and its frequency was set			
8	Planning and Communication			
8.1	The ALERT's Strategy map is clear and			
	understandable			
8.2	I understand the benefits of implementing			
	BSC			
8.3	BSC is better than previous measurement			
	systems			
8.4	BSC is relevant performance management			
	tool for the ALERT			
8.5	It is the right time for the ALERT to			
	implement BSC			

APPENDIX-II

Interview Questions for managers

- 1. Can you elaborate the major activities accomplished to properly put into practice the Balanced Scorecard?
- 2. Is enough training provided about BSC implementation and do employees have appropriate awareness about BSC?
- 3. Is the BSC system improved continuously based on a periodic assessment?
- 4. What are the challenges faced in the Balanced Scorecard measurement/evaluation system?
- 5. What measures should be taken to overcome the above problem?
- 6. Did BSC solve the challenges of the previous performance measurement tools?
- 7. Is there enough resource available to achieve the target if not what kind of action will do to improve this?
- 8. What improvements can be recommended concerning the BSC implementation process in the enterprise department?